Author's response to reviews

Title: How can continuing professional development better promote shared decision-making? Perspectives from an international collaboration

Authors:

France Légaré (france.legare@mfa.ulaval.ca)
Hilary Bekker (h.l.bekker@leeds.ac.uk)
Sophie Desroches (sophie.desroches@fsaa.ulaval.ca)
Renée Drolet (renee.drolet@crchuq.ulaval.ca)
Mary C Politi (mpoliti@wustl.edu)
Dawn Stacey (Dawn.Stacey@uOttawa.ca)
Francine Borduas (francineborduas@videotron.ca)
Francine M Cheater (Francine.Cheater@gcu.ac.uk)
Jacques Cornuz (Jacques.Cornuz@chuv.ch)
Marie-France Coutu (Marie-France.Coutu@USherbrooke.ca)
Nora Ferdjaoui-Moumjid (moumjid@univ-lyon1.fr)
Frances Griffiths (F.E.Griffiths@warwick.ac.uk)
Martin Härter (m.haerter@uke.de)
André Jacques (jacques@cmq.org)
Tanja Krones (tanja.krones@usz.ch)
Michel Labrecque (Michel.Labrecque@fmed.ulaval.ca)
Claire Neely (claireeneely.icsi@gmail.com)
Charo Rodriguez (charo.rodriguez@mcgill.ca)
Joan Sargeant (Joan.Sargeant@dal.ca)
Janet S Schuerman (janet.schuerman@icsi.org)
Mark D Sullivan (sullimar@u.washington.edu)

Version: 4 Date: 24 June 2011

Author's response to reviews:

Dear Editorial Team

we accepted all proposed copy editing changes. We also have reviewed email addresses and work addresses of each one of the authors. We have standardised the word "shared decision-making" because sometimes it was not using a ".".

thank you very much

France Légaré